Author's response to reviews

Title: A matched-pair cluster design study protocol to evaluate implementation of the Canadian C-Spine Rule in hospital emergency departments: phase III [NCT00290875]

Authors:

Ian G Stiell (istiell@ohri.ca)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
George A Wells (gawells@ottawaheart.ca)
Doug Coyle (dcoyle@uottawa.ca)
Howard J Lesiuk (hlesiuk@ottawahospital.on.ca)
Brian H Rowe (Brian Rowe@ualberta.ca)
Robert J Brison (brisonr@kgh.kari.net)
Michael John Schull (mjs@ices.on.ca)
Jacques Lee (jacques.lee@ices.on.ca)
Catherine M Clement (cclement@ohri.ca)

Version: 5 Date: 7 February 2007

Author's response to reviews:

Dr. Ian Stiell
Senior Scientist
Ottawa Health Research Institute
Ottawa Hospital, Civic Campus
1053 Carling Avenue, RM F657
Ottawa, ON K1Y 4E9

February 7, 2007

Martin Eccles & Brian Mittman
Editors-In-Chief
Implementation Science
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK

Dear Mr. Eccles and Mr. Mittman,

Please find attached the revised manuscript titled, "A matched-pair cluster design study protocol to evaluate implementation of the Canadian C-Spine Rule in hospital emergency departments: phase III [NCT00290875]."

Pursuant to The Biomed Central Editorial Production Team's email sent January 31st, 2007, the following revisions have been made to the manuscript:

* Trial registration number added to abstract
* Reference list contains all authors' names
* Figure legends and additional files listed after reference list as per examples on page 33
* All three word document figures have been cropped

Sincerely,

Ian G. Stiell MD MSc FRCPC

Attachment